Author's response to reviews

Title: LEARN 2 MOVE 7-12 years: a randomized controlled trial on the effects of a physical activity stimulation program in children with cerebral palsy

Authors:

Leontien Van Wely MSc (l.vanwely@vumc.nl)
Jules G Becher MD,PhD (jg.becher@vumc.nl)
Heleen A Reinders-Messelink PhD (H.A.Reinders@revalidatie-friesland.nl)
Eline Lindeman MD,PhD (E.Lindeman@umcutrecht.nl)
Olaf Verschuren PhD (o.verschuren@dehoogstraat.nl)
Johannes Verheijden (johannes@bosk.nl)
Annet J Dallmeijer PhD (A.Dallmeijer@vumc.nl)

Version: 3 Date: 1 November 2010

Author's response to reviews:

Please find enclosed a revised version of the study protocol of the LEARN 2 MOVE 7-12 study.

The following revisions have been made:

Main text
- All underlining has been removed and replaced by italic headings or extra paragraphs.
- Some typographical changes were made (spacing, returns)
- The lay-out of the manuscript has been adapted according to the lay-out mentioned in the biomedcentral “guideline for authors”.

Abstract
- A discussion section has been added
- One sentence has been removed in the Methods/Design section to gain extra space for the discussion. Removed sentence: (twice a week in the first two months, and then once a week in the following two months).

Methods
- The methods-section has been renamed as Methods/Design
- One paragraph has been rewritten (page 15, lines 12-20)
- The structure of the Paragraph "Outcome Measures" has been changed
- The abbreviation of StepWatch Activity Monitor has been changed in StepWatch, instead of SAM.
References
- The full citation details for any incomplete references have been provided for citations: 4, 5, 6, 10, 38, 41, 43, 51, 55.

Figures
- Both figures (1 and 2) have been cropped as closely as possible.